Author’s response to reviews

**Title:** Glucose control in intensive care: Usability, efficacy and safety of Space GlucoseControl in two medical European intensive care units

**Authors:**

Karin Amrein (karin.amrein@yahoo.de)
Norman Kachel (norman.kachel@bbraun.com)
Heike Fries (heike.fries@bbraun.com)
Roman Hovorka (rh347@cam.ac.uk)
Thomas R Pieber (thomas.pieber@medunigraz.at)
Johannes Plank (johannes.plank@elisabethinen.at)
Urs Wenger (urs.wenger@usz.ch)
Barbara Lienhardt (barbara.lienhardt@spitalbuelach.ch)
Marco Maggiorini (klinmax@usz.uzh.ch)

**Version:** 2  **Date:** 24 June 2014

**Author’s response to reviews:** see over
Dear Editors,

Thank you for the opportunity to revise and resubmit our manuscript. We would also like to thank both reviewers for their thoughtful comments, we appreciate this open review.

We have addressed all queries and additional editorial requests in the separate document *Response to Reviewers* and outlined the various changes to the manuscript in a point by point manner, these are highlighted in yellow in the revised version.

We hope that the revised and improved version fulfils the strict criteria for publication in *BMC Endocrine Disorders*.

Sincerely,

Karin Amrein, MD, MSc, corresponding author

Medical University Graz, Department of Internal Medicine, Auenbruggerplatz 15, Austria, A-8036 Graz
Phone: +43-316-660-4951714, Fax:+43-316-385-13428
Email: karin.amrein@medunigraz.at; karin.amrein@yahoo.de